Author's response to reviews

Title: Total Nasal Resistance among Sasang Constitutional Types: A Population-Based Study in Korea

Authors:

Dae Wui Yoon (dw1003@hotmail.com)
Seung Ku Lee (leeseungku@gmail.com)
Hyeryeon Yi (yihr@kornu.ac.kr)
Jeong Hwa Hong (hongjh32@korea.ac.kr)
Miyazaki Soichiro (soichiro@belle.shiga-med.ac.jp)
Si Woo Lee (ifree72@gmail.com)
Jong Yeol Kim (ssmed@kiom.re.kr)
Chol Shin (chol-shin@korea.ac.kr)

Version: 3 Date: 22 February 2013

Author's response to reviews:

Author’s covering letter for required changes to the initially submitted manuscript

Title: Total nasal resistance among sasang constitutional types: A population-based study in Korea

Date: 22 December 2013

Comments
I apologize for the inconvenience I am causing you. Requirements you asked have been changed appropriately as follows:

Requirements 1
Whether approval of an ethics committee was reported in the manuscript.

Answer: approval of an ethics committee for this study was given in the Methods section (line 4-5, page 6).

Requirements 2
Whether written informed consent for participation in the study was obtained from participants

Answer: sentence on written informed consent was inserted in the Methods section (line 21-22, page 5)

Requirements 3
Conforms your revised manuscript to the journal style

Answer: Affiliation in title page, size and style of title or sub-title has been
conformed to the BMC complementary and alternative medicine style.

Best regards.
Dae Wui Yoon.